Author's response to reviews

Title: Prevalence of lymphedema in women under treatment for breast cancer in a referral center in southeastern brazil

Authors:

Daniella MF Paiva (jcndani@yahoo.com.br)
Vivian O Rodrigues (viviolliver@hotmail.com)
Marcelle G Cesca (marcelle.goldner@gmail.com)
Pamella V Palma (pamellavalente@hotmail.com)
Isabel CG Leite (icgleite@hotmail.com)

Version: 5 Date: 9 November 2012

Author's response to reviews:

In response to the reviewer Aase S, A Sagen

1. The english was revised and the hypothesis was reformulated

We are at disposal.

The authors